```
Shepherd Andrew (Orcid ID: 0000-0002-4914-1299)
Gilbert Lin (Orcid ID: 0000-0002-4863-9589)
Muir Alan (Orcid ID: 0000-0001-7754-088X)
Konrad Hannes (Orcid ID: 0000-0002-5058-8637)
McMillan Malcolm (Orcid ID: 0000-0002-5113-0177)
Slater Thomas (Orcid ID: 0000-0003-2541-7788)
Briggs Kate Hannah (Orcid ID: 0000-0002-9831-7262)
Hogg Anna E (Orcid ID: 0000-0002-6441-4937)
```

Trends in Antarctic Ice Sheet Elevation and Mass

Andrew Shepherd¹, Lin Gilbert², Alan S. Muir^{2,3}, Hannes Konrad¹, Malcolm McMillan¹, Thomas Slater¹, Kate H. Briggs¹, Aud V. Sundal¹, Anna E. Hogg¹, Marcus Engdahl⁴

¹ Centre for Polar Observation and Modelling, School of Earth and Environment, University of Leeds, LS2 9JT, UK

² Mullard Space Science Laboratory, Department of Space & Climate Physics, University College London, WC1E 6BT, UK

³ Centre for Polar Observation and Modelling, Department of Earth Sciences, University College London, WC1E 6BT, UK

⁴ ESA-ESRIN, Via Galileo Galilei, Frascati (Roma), I-00044, Italy

Corresponding author: Andrew Shepherd (<u>a.shepherd@leeds.ac.uk</u>)

Key Points:

- We combine satellite radar altimetry and a regional climate model to separate changes in Antarctic elevation into snow and ice contributions
- Between 1992 and 2017, ice thinning has grown in extent to include 24% of West Antarctica, and reaches 122 metres in places
- Ice losses from Pine Island and Thwaites Glaciers have risen fivefold. East and West Antarctica have contributed 4.6±1.2 mm to sea level

Abstract

Fluctuations in Antarctic Ice Sheet elevation and mass occur over a variety of timescales, owing to changes in snowfall and ice flow. Here, we disentangle these signals by combining

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2019GL082182

25 years of satellite radar altimeter observations and a regional climate model. From these measurements, patterns of change that are strongly associated with glaciological events emerge. While the majority of the ice sheet has remained stable, 24% of West Antarctica is now in a state of dynamical imbalance. Thinning of the Pine Island and Thwaites glacier basins reaches 122 m in places, and their rates of ice loss are now five times greater than at the start of our survey. By partitioning elevation changes into areas of snow and ice variability, we estimate that East and West Antarctica have contributed -1.1 \pm 0.4 and +5.7 \pm 0.8 mm to global sea level between 1992 and 2017.

Plain Language Summary

Climate change can trigger long-term loss in ice sheet mass, which in turn drive global sea level rise. To isolate this signal, short-term fluctuations in snowfall have to be accounted for in measurements of ice sheet elevation change. In this study, we do this by combining satellite altimetry and a regional climate model. This allows us to show that the extent of ice sheet dynamical imbalance has grown over the past 25 years to incorporate 24% of the West Antarctic Ice Sheet. The thinning of Antarctic glaciers reaches 122 metres in places, and their combined ice losses have increased by a factor five. Altogether, Antarctica has contributed 4.6 ± 1.2 mm to global sea level rise.

1. Introduction

Satellite observations have transformed our knowledge of the Antarctic Ice Sheet (AIS) (Vaughan et al., 2013), and are now an important observational constraint for numerical simulations of its response to future climate change (Shepherd and Nowicki, 2017). Today, three satellite-based techniques are used to chart AIS imbalance; measurements of surface elevation change (e.g. (Pritchard et al., 2009; Wingham et al., 1998)), of ice flow (e.g. (Joughin et al., 1999; Rignot et al., 2011; Scambos et al., 1992)), and of changing gravitational attraction (e.g. (Luthcke et al., 2013; Velicogna and Wahr, 2006)). Although these methods have different strengths and weaknesses, their estimates of ice sheet mass balance agree when common geographical regions, time intervals, and models of surface mass balance and glacial isostatic adjustment are used (Shepherd et al., 2018). When high-resolution satellite measurements are combined with regional climate models (e.g. (Bromwich et al., 2011; Melchior Van Wessem et al., 2018)) or information on ice flow, it is possible to identify signals of ice sheet imbalance that are meteorological (Boening et al., 2012; Davis et al., 2005) and ice dynamical (Joughin et al., 2002; Rignot et al., 2005; Scambos et al., 2004; Shepherd et al., 2001; Sutterley et al., 2014) in origin.

Since 1992, the ERS-1 (Wingham et al., 1998), ERS-2 (Davis and Ferguson, 2004; Zwally et al., 2005), ENVISAT (Flament and Remy, 2012; Wingham et al., 2006a; Zwally et al., 2015), and CryoSat-2 (Helm et al., 2014; McMillan et al., 2014) satellite radar altimeters have measured changes in the AIS surface elevation at approximately monthly intervals. Between 2003 and 2009, the ICESat satellite laser altimeter also measured elevation changes during 18 campaigns (Pritchard et al., 2009; Zwally et al., 2015). The radar altimeter data constitute the longest continuous record of ice sheet wide change recorded by similar sensors with similar spatial and temporal sampling, and are a unique resource for studying ice sheet imbalance. In this paper we combine their measurements to determine changes in the elevation and volume of the East Antarctic (EAIS) and West Antarctic (WAIS) ice sheets, and parts of the Antarctic Peninsula ice sheet (APIS), over a 25-year period. We examine local and regional trends within the principal drainage basins of each ice sheet, and we provide an estimate of change within areas that are only partially surveyed. Using a firn densification model (Ligtenberg et al., 2011) driven by a regional climate model (Melchior Van Wessem et al., 2018) we then adjust the measured elevation changes to account for signals associated with fluctuations in snowfall, which allows us to locate and chart the evolution of ice sheet dynamical imbalance.

2. Ice Sheet Elevation Change

We developed time-series of surface elevation change across the AIS and within its principal drainage units from ERS-1, ERS-2, ENVISAT, and CryoSat-2 radar altimeter observations recorded between May 1992 and May 2017. Over 800 million altimeter measurements were included in our analysis, sampling 86, 76 and 79% of the APIS, EAIS and WAIS, respectively. To compute elevation, the altimeter range measurements were corrected for the lag of the leading-edge tracker, dry atmospheric mass, water vapour, the ionosphere, ocean loading tide, and temporal variations in surface scattering. We also adjusted for elevation changes associated with glacial isostatic adjustment using the IJ05_R2 model (Ivins et al., 2013). The surface elevation change time-series were computed (McMillan et al., 2014; Zwally et al., 1989) within regularly-spaced grid cells and during fixed time intervals (epochs). In the first instance we generated time-series from each satellite mission independently, and we then combined them by adjusting for the bias occurring during periods of mission overlap.

Although the majority of the grounded ice sheet was surveyed, a proportion fell beyond the southern limit of the satellite orbits (1% for CryoSat-2 and 21% for other missions). In other areas, where the altimeters failed to track rugged terrain (<2%) and in gaps between the satellite ground tracks, we interpolated the observed elevation changes using a triangulation scheme (Shepherd et al., 2001) at each epoch. We then computed elevation trends over discrete time intervals in each grid cell, within ice sheet drainage basins (Zwally et al., 2012) and also within the limits of the EAIS and the WAIS. We did not compute a regional assessment for the APIS because its northern drainage basins were too sparsely sampled. Finally, we modelled rates of elevation change in the remaining (mainly coastal) regions using local empirically-determined relationships with rates of ice flow.

To estimate the uncertainty of our elevation-change time series, we summed systematic and time-varying errors at each epoch, plus errors arising from the bias adjustment we apply to align each satellite mission. We estimated systematic uncertainties as the standard error of the long-term rate of elevation change in each region. Time-varying uncertainties at each epoch were computed as the average standard error of elevation measurements within the aggregated pixels. Uncertainties associated with the inter-mission bias were estimated as the standard deviation of differences between modelled elevations during a 2-year period centred on each mission overlap. At each epoch, uncertainties from all three error sources were summed in quadrature to estimate the overall error.

Alternative approaches have been proposed for certain stages in the processing of ice sheet elevation change. Key examples include (i) the method of determining the satellite range, which has been done with a variety of waveform retracking routines (e.g. (Davis, 1997; Helm et al., 2014; Nilsson et al., 2016; Wingham et al., 2004)) and with a variety of adjustments to account for temporally-correlated fluctuations in the radar echo shape (e.g. (Davis and Ferguson, 2004; Flament and Remy, 2012; Wingham et al., 1998; Zwally et al., 2005)), and (ii) the approach to forming elevation change time series, which has been done at orbit crossing points (Wingham et al., 1998; Zwally et al., 1989), along repeated ground tracks (Flament and Remy, 2012; Pritchard et al., 2009; Sorensen et al., 2011), and within fixed geographical regions (Helm et al., 2014; McMillan et al., 2014). While no approach has led to substantially different patterns of ice sheet elevation change in Antarctica, we nevertheless investigated the impact of many by testing alternative scenarios that employ

different approaches to (i) form the initial time-series, (ii) correct for correlated trends in echo shape, and to (iii) account for spatial and temporal omission.

To evaluate the processing scenarios (Figure 1), we considered their ability to match independent estimates of elevation change derived from precise airborne laser altimetry, and to address under-sampling of signal at northerly latitudes where the satellite orbit ground tracks diverge. Against these benchmarks, the optimal scenario among those we have considered uses a plane fit elevation change solver, averages measurements in 140-day epochs, interpolates elevation changes over distances of 20 km to fill remaining gaps, uses a 60-month period as the basis of corrections to compensate for correlated fluctuations in elevation and backscattered power, and does not apply a leading edge width correction. When compared to 544,422 repeat airborne laser altimetry measurements (Studinger, 2014), the optimal elevation change solution differed by 0.7 cm/year, on average, with CryoSat-2's interferometric mode showing 15 % lower variance thanks to its smaller ground footprint and ability to more precisely locate the angle to the surface reflection. The average difference is smaller than the estimated certainty of our 25-year rate of elevation change for WAIS (1.0 cm/yr) – an area of comparable size - which suggests that our error model adequately captures the principal sources of uncertainty. Moreover, of the 90 alternative scenarios we considered, 86% and 83% produced EAIS and WAIS elevation trends that fell within the estimated 1-sigma (67%) uncertainty of our optimal solution, respectively, which suggests that our error model also adequately captures variance associated with the processing methods we have tested. For more details on the altimetry methods see the supporting information.

Although most of the AIS surface has changed little in elevation over the past 25 years (Figure 2), there are clear patterns of thinning and thickening in coastal sectors especially in WAIS and the APIS. Some of these signals have been identified in shorter records (Flament and Remy, 2012; McMillan et al., 2014; Pritchard et al., 2009; Shepherd and Wingham, 2007; Shepherd et al., 2002; Wingham et al., 1998; Zwally et al., 2005) and are now better defined because our time series is long in comparison to the period of snowfall fluctuations, which are typically decadal or less in Antarctica (Shepherd et al., 2012; Wouters et al., 2013). Most of the significant changes are coincident with glaciers and ice streams (Rignot et al., 2011). In some cases, the origins of these signals have been related to changes in ice flow through independent observations, for example slowdown of glaciers at the Siple Coast (Anandakrishnan and Alley, 1997)(Scheuchl et al., 2012) is driving ice sheet thickening in this sector, and speedup of glaciers draining into the Amundsen Sea (Joughin et al., 2003; Mouginot et al., 2014) and of Fleming Glacier (Rignot et al., 2005) is causing thinning. In other cases, ice sheet thinning has been related to dynamical imbalance through indirect measures such as excessive ice discharge or grounding line retreat; Totten Glacier (Li et al., 2015) and several Bellingshausen Sea glaciers (Christie et al., 2016) fall into this category, as a contemporary speedup has not been recorded.

Elsewhere, there are signals of ice sheet elevation change that are distal to areas of fast ice flow. In Dronning Maud Land (basins 5 to 8), a broad pattern of modest ice sheet thickening spans much of the coastline and stretches several hundred kilometres inland; this has been associated with sharp increases in snowfall that occurred between 2009 and 2012 (Boening et al., 2012). Modest thickening of the ice sheet to the west of Ronne Ice Shelf (basin 1) may also be meteorological in origin. The magnitudes of these signals are small (5 to 10 cm/yr), and do not coincide with areas of rapid ice flow or of known ice dynamical imbalance. However, the same cannot be said for stronger patterns of ice sheet thickening in the upper reaches of drainage catchments of the Bellingshausen Sea sector (basins 23 and 24), or of ice sheet thinning inland of the southern Siple Coast ice streams (basins 17 and 18);

these changes are adjacent to areas where ice flow units are changing in thickness and complicate their interpretation. In the southwestern Antarctic Peninsula, for example, a modest speedup of glaciers flowing into George VI Ice Shelf accounts for only a fraction of the observed coastal deflation (Hogg et al., 2017) and, similarly, thinning of glaciers flowing into the Getz Ice Shelf is greater than estimated increases in ice discharge alone (Chuter et al., 2017). In these two sectors, the pattern of elevation change reflects a complex mix of surface processes and ice dynamical imbalance.

3. Ice Dynamical Imbalance

Changes in Antarctic ice sheet elevation arise predominantly due to fluctuations in accumulation and ice flow, which occur at the densities of snow and ice, respectively (Wingham, 2000). These processes can be distinguishable in altimeter records due to their coincidence with areas of rapid or changing ice flow, or due to their persistence over periods that are long in comparison to expected snowfall fluctuations. To discriminate them, we adjusted the satellite elevation changes to account for fluctuations in surface mass balance by removing model estimates of the firn layer thickness change (Ligtenberg et al., 2011; Melchior Van Wessem et al., 2018). We then classified regions exhibiting correlated patterns of sustained and significant thickening or thinning relative to the firn thickness changes as areas of ice dynamical imbalance. In some places, ice dynamical imbalance has spread inland (e.g. (Konrad et al., 2017)), and so we allowed these areas to grow over time if a significant accelerated elevation trend is present. Using this approach there is, for example, a ~200,000 km² increase in the area of dynamical imbalance in the Amundsen Sea sector over the 25year survey period, which corresponds to a 150 to 300 m/yr decrease in the average speed (Rignot et al., 2011) of ice that is thinning. Altogether, 56,950 km², 415,175 km², and 17,900 km² of the East Antarctic, West Antarctic, and Antarctic Peninsula ice sheets were identified in this scheme to be in a state of dynamical imbalance by the end of 2017. Elsewhere, we assumed that elevation changes were caused by fluctuations in surface mass balance, though potential signals of dynamical imbalance that are modest, highly localised, abrupt, or episodic, may not be fully resolved. For more details on the ice dynamical imbalance methods see the supporting information.

Our classification scheme identifies long-term ice dynamical imbalance in ten drainage basins, including two where the area of imbalance grows over time. At the Siple Coast, ice thickening is steady and uniform across three southern ice flow units which have inflated by 232±27 cm, on average, over the 25-year survey. In contrast, thinning of the Pine Island, Thwaites, and Totten glaciers increases towards the ice sheet margin, peaking at 80±2, 122±1, and 26±1 m, respectively, at their termini. These glaciers drain the principal marinebased and ocean-terminating sectors of Antarctica, a geometrical configuration that is theoretically unstable (Schoof, 2007) and is in numerical simulations highly-sensitive to ice marginal perturbations (Joughin et al., 2014; Payne et al., 2004). There is also evidence of ocean-driven melting at their termini (Dutrieux et al., 2014; Jacobs et al., 2011; Rintoul et al., 2016). On average, the sections of the Pine Island, Thwaites, and Totten glaciers identified to be in a state of dynamical imbalance thinned at rates of 45 ± 5 , 49 ± 5 , and 25 ± 3 cm/yr, respectively, between 1992 and 2017, whereas the remainder of their catchments were relatively stable. At the Totten glacier, dynamical imbalance affects just 3.6% of the drainage basin, and we found little evidence that either this or the rate of elevation change have changed over the 25-year survey. In contrast, ice drawdown within the Pine Island and Thwaites glacier drainage basins has spread rapidly inland, and now affects the majority (51% and 68%, respectively) of their catchments.

4. Ice Sheet Mass Balance

We used our classification of ice dynamical imbalance as the basis of a spatiallyresolved mass balance calculation, as it allows us to separate elevation changes predominantly occurring at the densities of snow and ice. Our classification is an improvement over previous schemes because it uses model estimates of firn thickness change to locate the change in ice thickness, rather than attributing elevation changes within entire drainage sectors (Wingham et al., 1998) or otherwise defined regions (Shepherd et al., 2012) to ice. It does not, however, account for potentially coincident signals of meteorological and dynamical imbalance, and this is an acknowledged shortcoming. Nevertheless, although making an explicit, model-based correction for snowfall fluctuations (Zwally et al., 2015) is in principal ideal, our classification scheme remains preferable because (i) firn models are not well matched with satellite radar altimeter elevation changes, because (ii) their application leads to spurious signals of ice dynamical imbalance in areas changing at rates below the certainty of the altimeter and firn model trends, and because (iii) the standard deviation between the resulting mass trends is 43% lower when compared to independent estimates derived from satellite gravimetry (Table S4). We instead treated snowfall as an additional source of elevation change uncertainty in mass balance calculations by summing an estimate of its variability (Table 1) in quadrature with the satellite elevation trend uncertainties. For more detailed on the mass balance methods and evaluation see the supporting information.

Seven drainage basins are situated partly within the region that falls beyond the satellites' southern orbital limits. Although this area represents a sizeable fraction (21%) of the continental ice sheet for ERS-1, ERS-2, and ENVISAT, the broader pattern of elevation change within the Antarctic interior has remained stable over the survey period, and is consistent with the changes recorded farther south by CryoSat-2 since 2010; i.e. the Kamb, Whillans, and Mercer ice streams have thickened, steadily, while little change has occurred elsewhere. Our estimate of the ice sheet interior mass balance is based on elevation trends recorded in the southernmost degree of latitude sampled, and puts the region in a state close to balance – gaining 1.7±0.1 Gt/yr, on average, since 1992. At the continental scale, we estimate that the EAIS and WAIS changed in mass at average rates of +16±6 and -82±12 Gt/yr, respectively, between 1992 and 2017 (Table 1). These values are in close agreement with a recent community assessment of 24 satellite altimetry, gravimetry, and mass budget surveys (Shepherd et al., 2018), which put the EAIS and WAIS mass trends at +5±46 and -94±27 Gt/yr over the same period. Our estimates of ice sheet mass balance are also in close agreement with estimates (Groh and horwath, 2016) derived from GRACE satellite gravimetry; across the 23 drainage basins included in our survey and over a common 13-year period (2002 to 2016), the root mean square difference between the two estimates is 4.9 Gt/yr.

By far the largest signal of imbalance has occurred in the Pine Island and Thwaites glacier drainage basins (Figure 3), which have lost mass at average rates of -28 ± 6 Gt/yr and -46 ± 7 Gt/yr, respectively, since 1992 (Table 1). Despite a steady gain of 13 ± 4 Gt/yr within the southern Siple Coast ice streams (Table 2), Amundsen Sea sector glaciers dominate the overall mass balance of WAIS which has lost an estimated $1,851\pm133$ Gt of ice over the 25-year survey. Although we have identified an area of persistent thinning at Totten glacier as being dynamical in origin, these losses (9 ± 1 Gt/yr) are small by comparison to changes in WAIS. They are also offset by modest ice sheet thickening across the remainder of the EAIS which, overall, has gained an estimated 407 ± 161 Gt of snow and ice since 1992.

Accelerating mass loss from the Amundsen Sea Sector has been documented in a wide range of satellite altimetry (Thomas et al., 2004; Wingham et al., 2009), mass budget

(Medley et al., 2014; Rignot et al., 2008), and gravimetry (Bouman et al., 2014; Velicogna et al., 2014) surveys, and in comparisons the techniques have shown close agreement (Shepherd et al., 2012; Sutterley et al., 2014). We examined this signal by calculating mass trends within the regions of ice dynamical imbalance of the Pine Island and Thwaites glacier drainage basins over 5-year intervals (see Table S3). For comparison, we repeated this calculation also for the Totten and Siple Coast glaciers, where the area of dynamical imbalance has remained static. Rates of ice mass loss from the Pine Island and Thwaites glaciers have increased progressively over time, rising from 2 ± 1 and 12 ± 1 Gt/yr, respectively, between 1992 and 1997, to 55 ± 4 and 76 ± 6 Gt/yr, respectively, during the latter period of our survey (2012 to 2016). A consequence of this inland spreading is that over half (59%) of the Amundsen Sea sector is now in a state of dynamical imbalance. In keeping with the findings of previous studies (Medley et al., 2014; Rignot et al., 2008; Rignot et al., 2014), our data confirm that the Pine Island and Thwaites Glacier are by far the largest regional contributors to global sea level rise. In contrast, rates of mass loss and gain from the Totten and Siple Coast glaciers have remained relatively stable over time.

5 Conclusions

Our ERS-1, ERS-2, ENVISAT and CryoSat-2 radar altimetry time-series provides a comprehensive assessment of the spatial and temporal pattern of AIS elevation change between 1992 and 2017. When combined with model estimates of firn thickness change (Ligtenberg et al., 2011; Melchior Van Wessem et al., 2018), the 25-year altimeter record enables long-term trends in ice thickness to be separated from short-term snowfall variability. While the majority of the ice sheet surface has remained stable over the survey period, areas of ice dynamical imbalance are now clearly apparent across many sectors of the continent. We estimate that 0.6%, 23.7%, and 7.9% of the EAIS, WAIS, and APIS ice sheets are now in a state of dynamical imbalance. Assuming changes in these areas have occurred at the density of ice, and the remainder at the density of snow, we estimate that, together, the EAIS and WAIS changed in mass at an average rate of -66±18 Gt/yr between 1992 and 2017. This equates to an average 4.6±1.2 mm contribution to global sea level over the same period, a value that is comparable to estimates determined using other approaches (The IMBIE Team, 2018). Ice losses from the Amundsen Sea sector of WAIS have increased progressively over the course of our survey, and were five times greater in the final decade of our survey than during the initial decade. A future goal is to develop an optimal altimetry ice sheet mass balance solution that fully incorporates model estimates of firn thickness and surface mass balance.

Acknowledgments and Data

We thank Michiel van den Broeke and Stefan Ligtenberg for providing regional climate model firn density model output. GRACE data were sourced from https://data1.geo.tu-dresden.de/ais_gmb on 12/08/16. IceBridge data were sourced from https://n5eil01u.ecs.nsidc.org/ICEBRIDGE/IDHDT4.001 on 25/05/17. From the date of publication, the ice sheet elevation and mass changes reported in this study will be available at http://www.cpom.ucl.ac.uk/csopr/





Anandakrishnan, S. and Alley, R. B.: Stagnation of Ice Stream C, West Antarctica by water piracy, Geophysical Research Letters, 24, 265-268, 1997.

Arthern, R. J. and Wingham, D. J.: The natural fluctuations of firn densification and their effect on the geodetic determination of ice sheet mass balance, Climatic Change, 40, 605-624, 1998.

Bamber, J. L.: Ice-Sheet Altimeter Processing Scheme, International Journal of Remote Sensing, 15, 925-938, 1994.

Boening, C., Lebsock, M., Landerer, F., and Stephens, G.: Snowfall-driven mass change on the East Antarctic ice sheet, Geophysical Research Letters, 39, 2012.

Bouman, J., Fuchs, M., Ivins, E., Van Der Wal, W., Schrama, E., Visser, P., and Horwath, M.: Antarctic outlet glacier mass change resolved at basin scale from satellite gravity gradiometry, Geophysical Research Letters, 41, 5919-5926, 2014.

Bromwich, D. H., Nicolas, J. P., and Monaghan, A. J.: An Assessment of precipitation changes over antarctica and the southern ocean since 1989 in contemporary global reanalyses, Journal of Climate, 24, 4189-4209, 2011.

Brown, G. S.: The average impulse response of a rough surface and its applications, IEEE Transactions of Antennas and Propogation, 25, 67-74, 1977.

Christie, F. D. W., Bingham, R. G., Gourmelen, N., Tett, S. F. B., and Muto, A.: Four-decade record of pervasive grounding line retreat along the Bellingshausen margin of West Antarctica, Geophysical Research Letters, 43, 5741-5749, 2016.

Chuter, S. J., Martín-Español, A., Wouters, B., and Bamber, J. L.: Mass balance reassessment of glaciers draining into the Abbot and Getz Ice Shelves of West Antarctica, Geophysical Research Letters, 44, 7328-7337, 2017.

Cornford, S. L., Martin, D. F., Graves, D. T., Ranken, D. F., Le Brocq, A. M., Gladstone, R. M., Payne, A. J., Ng, E. G., and Lipscomb, W. H.: Adaptive mesh, finite volume modeling of marine ice sheets, J. Comput. Phys., 232, 529-549, 2013.

Davis, C. H.: A robust threshold retracking algorithm for measuring ice-sheet surface elevation change from satellite radar altimeters, IEEE Transactions on Geoscience and Remote Sensing, 35, 974-979, 1997.

Davis, C. H. and Ferguson, A. C.: Elevation change of the Antarctic ice sheet, 1995-2000, from ERS-2 satellite radar altimetry, Ieee Transactions on Geoscience and Remote Sensing, 42, 2437-2445, 2004.

Davis, C. H., Li, Y., McConnell, J. R., Frey, M. M., and Hanna, E.: Snowfall-Driven Growth in East Antarctic Ice Sheet Mitigates Recent Sea-Level Rise, Science, 308 1898-190,12005.

Dutrieux, P., De Rydt, J., Jenkins, A., Holland, P. R., Ha, H. K., Lee, S. H., Steig, E. J., Ding, Q., Abrahamsen, E. P., and Schröder, M.: Strong sensitivity of pine Island ice-shelf melting to climatic variability, Science, 343, 174-178, 2014.

Flament, T. and Remy, F.: Antarctica volume change from 10 years of Envisat altimetry, in: International Geoscience and Remote Sensing Symposium (IGARSS)2012, 1848-1851.

Groh, A. and horwath, M.: The method of tailored sensitivity kernels for GRACE mass change estimates, European Geophysical Union, Vienna, 2016.

Helm, V., Humbert, A., and Miller, H.: Elevation and elevation change of Greenland and Antarctica derived from CryoSat-2, Cryosphere, 8, 1539-1559, 2014.

Hogg, A. E., Shepherd, A., Cornford, S. L., Briggs, K. H., Gourmelen, N., Graham, J. A., Joughin, I., Mouginot, J., Nagler, T., Payne, A. J., Rignot, E., and Wuite, J.: Increased ice flow in Western Palmer Land linked to ocean melting, Geophysical Research Letters, 44, 4159-4167, 2017.

Horwath, M., Legrésy, B., Rémy, F., Blarel, F., and Lemoine, J. M.: Consistent patterns of Antarctic ice sheet interannual variations from ENVISAT radar altimetry and GRACE satellite gravimetry, Geophysical Journal International, 189, 863-876, 2012.

Iijima, B. A., Harris, I. L., Ho, C. M., Lindqwister, U. J., Mannucci, A. J., Pi, X., Reyes, M. J., Sparks, L. C., and Wilson, B. D.: Automated daily process for global ionospheric total electron content maps and satellite ocean altimeter ionospheric calibration based on Global Positioning System data, Journal of Atmospheric and Solar-Terrestrial Physics, 61, 1205-1218, 1999.

Ivins, E. R., James, T. S., Wahr, J., O. Schrama, E. J., Landerer, F. W., and Simon, K. M.: Antarctic contribution to sea level rise observed by GRACE with improved GIA correction, J. Geophys. Res. B Solid Earth, 118, 3126-3141, 2013.

Jacobs, S. S., Jenkins, A., Giulivi, C. F., and Dutrieux, P.: Stronger ocean circulation and increased melting under Pine Island Glacier ice shelf, Nature Geoscience, 4, 519-523, 2011.

Joughin, I., Rignot, E., Rosanova, C. E., Lucchitta, B. K., and Bohlander, J.: Timing of recent accelerations of Pine Island Glacier, Antarctica, Geophysical Research Letters, 30, 1706, 2003.

Joughin, I., Smith, B. E., and Medley, B.: Marine ice sheet collapse potentially under way for the thwaites glacier basin, West Antarctica, Science, 344, 735-738, 2014.

Joughin, I., Tulaczyk, S., Bindschadler, R., and Price, S. F.: Changes in west Antarctic ice stream velocities: Observation and analysis, J. Geophys. Res. B Solid Earth, 107, EPM 3-1 - 3-22, 2002.

Joughin, L., Gray, L., Bindschadler, R., Price, S., Morse, D., Hulbe, C., Mattar, K., and Werner, C.: Tributaries of West Antarctic Ice streams revealed by RADARSAT interferometry, Science, 286, 283-286, 1999.

Khvorostovsky, K. S.: Merging and analysis of elevation time series over greenland ice sheet from satellite radar altimetry, IEEE Transactions on Geoscience and Remote Sensing, 50, 23-36, 2012.

Konrad, H., Gilbert, L., Cornford, S. L., Payne, A., Hogg, A., Muir, A., and Shepherd, A.: Uneven onset and pace of ice-dynamical imbalance in the Amundsen Sea Embayment, West Antarctica, Geophysical Research Letters, 44, 910-918, 2017.

Li, X., Rignot, E., Morlighem, M., Mouginot, J., and Scheuchl, B.: Grounding line retreat of Totten Glacier, East Antarctica, 1996 to 2013, Geophysical Research Letters, 42, 8049-8056, 2015.

Ligtenberg, S. R. M., Helsen, M. M., and Van Den Broeke, M. R.: An improved semiempirical model for the densification of Antarctic firn, Cryosphere, 5, 809-819, 2011.

Luthcke, S. B., Sabaka, T. J., Loomis, B. D., Arendt, A. A., McCarthy, J. J., and Camp, J.: Antarctica, Greenland and Gulf of Alaska land-ice evolution from an iterated GRACE global mascon solution, Journal of Glaciology, 59, 613-631, 2013.

McMillan, M., Shepherd, A., Sundal, A., Briggs, K., Muir, A., Ridout, A., Hogg, A., and Wingham, D.: Increased ice losses from Antarctica detected by CryoSat-2, Geophysical Research Letters, 41, 3899-3905, 2014.

Medley, B., Joughin, I., Smith, B. E., Das, S. B., Steig, E. J., Conway, H., Gogineni, S., Lewis, C., Criscitiello, A. S., McConnell, J. R., Van Den Broeke, M. R., Lenaerts, J. T. M., Bromwich, D. H., Nicolas, J. P., and Leuschen, C.: Constraining the recent mass balance of pine island and thwaites glaciers, west antarctica, with airborne observations of snow accumulation, Cryosphere, 8, 1375-1392, 2014.

Melchior Van Wessem, J., Jan Van De Berg, W., Noël, B. P. Y., Van Meijgaard, E., Amory, C., Birnbaum, G., Jakobs, C. L., Krüger, K., Lenaerts, J. T. M., Lhermitte, S., Ligtenberg, S. R. M., Medley, B., Reijmer, C. H., Van Tricht, K., Trusel, L. D., Van Ulft, L. H., Wouters, B., Wuite, J., and Van Den Broeke, M. R.: Modelling the climate and surface mass balance of polar ice sheets using RACMO2 - Part 2: Antarctica (1979-2016), Cryosphere, 12, 1479-1498, 2018.

Michel, A., Flament, T., and Rémy, F.: Study of the penetration bias of ENVISAT altimeter observations over Antarctica in comparison to ICESat observations, Remote Sens., 6, 9412-9434, 2014.

Mouginot, J., Rignot, E., and Scheuchl, B.: Sustained increase in ice discharge from the Amundsen Sea Embayment, West Antarctica, from 1973 to 2013, Geophysical Research Letters, 41, 1576-1584, 2014.

Nilsson, J., Gardner, A., Sørensen, L. S., and Forsberg, R.: Improved retrieval of land ice topography from CryoSat-2 data and its impact for volume-change estimation of the Greenland Ice Sheet, Cryosphere, 10, 2953-2969, 2016.

Payne, A. J., Vieli, A., Shepherd, A. P., Wingham, D. J., and Rignot, E.: Recent dramatic thinning of largest West Antarctic ice stream triggered by oceans, Geophysical Research Letters, 31, art-L23401, 2004.

Pritchard, H. D., Arthern, R. J., Vaughan, D. G., and Edwards, L. A.: Extensive dynamic thinning on the margins of the Greenland and Antarctic ice sheets, Nature, 461, 971-975, 2009.

Rignot, E., Bamber, J. L., Van Den Broeke, M. R., Davis, C., Li, Y., Van De Berg, W. J., and Van Meijgaard, E.: Recent Antarctic ice mass loss from radar interferometry and regional climate modelling, Nature Geoscience, 1, 106-110, 2008.

Rignot, E., Casassa, G., Gogineni, P., Krabill, W., Rivera, A., and Thomas, R.: Accelerated ice discharge from the Antarctic Peninsula following the collapse of Larsen B ice shelf, Geophysical Research Letters, 31, art-L18401, 2004.

Rignot, E., Casassa, G., Gogineni, S., Kanagaratnam, P., Krabill, W., Pritchard, H., Rivera, A., Thomas, R., Turner, J., and Vaughan, D.: Recent ice loss from the Fleming and other glaciers, Wordie Bay, West Antarctic Peninsula, Geophysical Research Letters, 32, 1-4, 2005.

Rignot, E., Mouginot, J., Morlighem, M., Seroussi, H., and Scheuchl, B.: Widespread, rapid grounding line retreat of Pine Island, Thwaites, Smith, and Kohler glaciers, West Antarctica, from 1992 to 2011, Geophysical Research Letters, 41, 3502-3509, 2014.

Rignot, E., Mouginot, J., and Scheuchl, B.: Ice Flow of the Antarctic Ice Sheet, Science, 333, 1427-1430, 2011.

Rintoul, S. R., Silvano, A., Pena-Molino, B., Van Wijk, E., Rosenberg, M., Greenbaum, J. S., and Blankenship, D. D.: Ocean heat drives rapid basal melt of the totten ice shelf, Sci. Adv., 2, 2016.

Scambos, T. A., Bohlander, J. A., Shuman, C. A., and Skvarca, P.: Glacier acceleration and thinning after ice shelf collapse in the Larsen B embayment, Antarctica, Geophysical Research Letters, 31, art-L18402, 2004.

Scambos, T. A., Dutkiewicz, M. J., Wilson, J. C., and Bindschadler, R. A.: Application of image crosscorrelation to the measurement of glacier velocity using satellite image data, Remote Sensing of the Environment, 42, 177-186, 1992.

Scheuchl, B., Mouginot, J., and Rignot, E.: Ice velocity changes in the Ross and Ronne sectors observed using satellite radar data from 1997 and 2009, Cryosphere, 6, 1019-1030, 2012.

Schoof, C.: Ice sheet grounding line dynamics: Steady states, stability, and hysteresis, Journal of Geophysical Research-Earth Surface, 112, 2007.

Shepherd, A., Ivins, E., Rignot, E., Smith, B., Van Den Broeke, M., Velicogna, I., Whitehouse, P., Briggs, K., Joughin, I., Krinner, G., Nowicki, S., Payne, T., Scambos, T., Schlegel, N., Geruo, A., Agosta, C., Ahlstrøm, A., Babonis, G., Barletta, V., Blazquez, A., Bonin, J., Csatho, B., Cullather, R., Felikson, D., Fettweis, X., Forsberg, R., Gallee, H., Gardner, A., Gilbert, L., Groh, A., Gunter, B., Hanna, E., Harig, C., Helm, V., Horvath, A., Horwath, M., Khan, S., Kjeldsen, K. K., Konrad, H., Langen, P., Lecavalier, B., Loomis, B., Luthcke, S., McMillan, M., Melini, D., Mernild, S., Mohajerani, Y., Moore, P., Mouginot, J., Moyano, G., Muir, A., Nagler, T., Nield, G., Nilsson, J., Noel, B., Otosaka, I., Pattle, M. E., Peltier, W. R., Pie, N., Rietbroek, R., Rott, H., Sandberg-Sørensen, L., Sasgen, I., Save, H.,

Scheuchl, B., Schrama, E., Schröder, L., Seo, K. W., Simonsen, S., Slater, T., Spada, G., Sutterley, T., Talpe, M., Tarasov, L., Van De Berg, W. J., Van Der Wal, W., Van Wessem, M., Vishwakarma, B. D., Wiese, D., and Wouters, B.: Mass balance of the Antarctic Ice Sheet from 1992 to 2017, Nature, 558, 219-222, 2018.

Shepherd, A., Ivins, E. R., Geruo, A., Barletta, V. R., Bentley, M. J., Bettadpur, S., Briggs, K. H., Bromwich, D. H., Forsberg, R., Galin, N., Horwath, M., Jacobs, S., Joughin, I., King, M. A., Lenaerts, J. T. M., Li, J., Ligtenberg, S. R. M., Luckman, A., Luthcke, S. B., McMillan, M., Meister, R., Milne, G., Mouginot, J., Muir, A., Nicolas, J. P., Paden, J., Payne, A. J., Pritchard, H., Rignot, E., Rott, H., Sørensen, L. S., Scambos, T. A., Scheuchl, B., Schrama, E. J. O., Smith, B., Sundal, A. V., Van Angelen, J. H., Van De Berg, W. J., Van Den Broeke, M. R., Vaughan, D. G., Velicogna, I., Wahr, J., Whitehouse, P. L., Wingham, D. J., Yi, D., Young, D., and Zwally, H. J.: A reconciled estimate of ice-sheet mass balance, Science, 338, 1183-1189, 2012.

Shepherd, A. and Nowicki, S.: Improvements in ice-sheet sea-level projections, Nat. Clim. Change, 7, 672-674, 2017.

Shepherd, A. and Wingham, D.: Recent sea-level contributions of the Antarctic and Greenland ice sheets, Science, 315, 1529-1532, 2007.

Shepherd, A., Wingham, D. J., and Mansley, J. A. D.: Inland thinning of the Amundsen Sea sector, West Antarctica, Geophysical Research Letters, 29, 2002.

Shepherd, A., Wingham, D. J., Mansley, J. A. D., and Corr, H. F. J.: Inland thinning of Pine Island Glacier, West Antarctica, Science, 291, 862-864, 2001.

Sorensen, L. S., Simonsen, S. B., Nielsen, K., Lucas-Picher, P., Spada, G., Adalgeirsdottir, G., Forsberg, R., and Hvidberg, C. S.: Mass balance of the Greenland ice sheet (2003-2008) from ICESat data - the impact of interpolation, sampling and firn density, Cryosphere, 5, 173-186, 2011.

Studinger, M.: IceBridge ATM L4 Surface Elevation Rate of Change, Version 299 1, Antarctica subset. Center, N. S. a. I. D. C. D. A. A. (Ed.), Boulder, Colorado, USA, 2014.

Sutterley, T. C., Velicogna, I., Rignot, E., Mouginot, J., Flament, T., Van Den Broeke, M. R., Van Wessem, J. M., and Reijmer, C. H.: Mass loss of the Amundsen Sea Embayment of West Antarctica from four independent techniques, Geophysical Research Letters, 41, 8421-8428, 2014.

The IMBIE Team, T. I.: Mass balance of the Antarctic ice sheet from 1992 to 2017, Nature, 558, 219-222, 2018.

Thomas, R., Rignot, E., Casassa, G., Kanagaratnam, P., Acuna, C., Akins, T., Brecher, H., Frederick, E., Gogineni, P., Krabill, W., Manizade, S., Ramamoorthy, H., Rivera, A., Russell, R., Sonntag, J., Swift, R., Yungel, J., and Zwally, J.: Accelerated sea-level rise from West Antarctica, Science, 306, 255-258, 2004.

Vaughan, D., comiso, J., allison, I., carrascho, J., kaser, G., Kwok, R., Mote, P., Murray, T., paul, F., Ren, J., Rignot, E., Solomina, O., Steffen, K., and Zhang, T.: Observations: Cryosphere. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change,

Stocker, T. F., Qin, D., Plattner, G.-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y., Bex, V., and Midgeley, P. M. (Eds.), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 2013.

Velicogna, I., Sutterley, T. C., and Van Den Broeke, M. R.: Regional acceleration in ice mass loss from Greenland and Antarctica using GRACE time-variable gravity data, Geophysical Research Letters, 41, 8130-8137, 2014.

Velicogna, I. and Wahr, J.: Measurements of time-variable gravity show mass loss in Antarctica, Science, 311, 1754-1756, 2006.

Wingham, D. J.: Small fluctuations in the density and thickness of a dry firn column, Journal of Glaciology, 46, 399-412, 2000.

Wingham, D. J., Francis, C. R., Baker, S., Bouzinac, C., Cullen, R., de Chateau-Thierry, P., Laxon, S. W., Mallow, U., Mavrocordatos, C., Phalippou, L., Ratier, G., Rey, L., Rostan, F., Viau, P., and Wallis, D.: CryoSat: A Mission to Determine the Fluctuations in Earth's Land and Marine Ice Fields, Advances in Space Research, 2004. 2004.

Wingham, D. J., Ridout, A. J., Scharroo, R., Arthern, R. J., and Shum, C. K.: Antarctic elevation change from 1992 to 1996, Science, 282, 456-458, 1998.

Wingham, D. J., Shepherd, A., Muir, A., and Marshall, G. J.: Mass balance of the Antarctic ice sheet, Philosophical Transactions of the Royal Society A-Mathematical Physical and Engineering Sciences, 364, 1627-1635, 2006a.

Wingham, D. J., Siegert, M. J., Shepherd, A., and Muir, A. S.: Rapid discharge connects Antarctic subglacial lakes, Nature, 440, 1033-1036, 2006b.

Wingham, D. J., Wallis, D. W., and Shepherd, A.: Spatial and temporal evolution of Pine Island Glacier thinning, 1995-2006, Geophysical Research Letters, 36, 2009.

Wouters, B., Bamber, J. L., Van Den Broeke, M. R., Lenaerts, J. T. M., and Sasgen, I.: Limits in detecting acceleration of ice sheet mass loss due to climate variability, Nature Geoscience, 6, 613-616, 2013.

Zwally, H. J., Brenner, A. C., Major, J. A., Bindschadler, R. A., and Marsh, J. G.: Growth of Greenland ice sheet: Measurement, Science, 246, 1587-1589, 1989.

Zwally, H. J., Giovinetto, M. B., Beckley, M. A., and Saba, J. L.: Antarctic and Greenland drainage systems. GSFC Cryospheric Sciences Laboratory, 2012.

Zwally, H. J., Giovinetto, M. B., Li, J., Cornejo, H. G., Beckley, M. A., Brenner, A. C., Saba, J. L., and Yi, D. H.: Mass changes of the Greenland and Antarctic ice sheets and shelves and contributions to sea-level rise: 1992-2002, Journal of Glaciology, 51, 509-527, 2005.

Zwally, H. J., Li, J., Robbins, J. W., Saba, J. L., Yi, D., and Brenner, A. C.: Mass gains of the Antarctic ice sheet exceed losses, Journal of Glaciology, 61, 1019-1036, 2015.

Table 1. The observed area, mean accumulation rate (MAR), estimated snowfall variability, average elevation rate, and average mass balance of AIS drainage basins between May 1992 and May 2017. Basins with regions of identified ice dynamical imbalance are italicised, and where elevation changes also exceed expected snowfall variability are in bold.

| Drainage basin | Area | Mean ice | Snowfall | Elevation | Average |
|----------------|--------------------|--------------|-------------|-----------------|-----------------|
| | (km ²) | accumulation | variability | rate | mass |
| | | rate | (cm/yr) | (cm/yr) | balance |
| | | (cm/yr) | | | (Gt/yr) |
| 1 | 465525 | 29 | 4.2 | 3.1 ± 0.3 | 5.7 ± 2.1 |
| 2 | 741575 | 7 | 1.6 | 0.3 ± 2.3 | 0.7 ± 6.2 |
| 3 | 1501700 | 5 | 0.4 | 0.7 ± 0.1 | 3.6 ± 0.6 |
| 4 | 239475 | 22 | 3.2 | 3.3 ± 0.4 | 3.1 ± 0.6 |
| 5 | 184750 | 21 | 4.3 | 4.4 ± 0.6 | 3.2 ± 0.6 |
| 6 | 599025 | 13 | 1.4 | 1.8 ± 0.3 | 4.1 ± 0.8 |
| 7 | 493875 | 18 | 2.3 | 1.6 ± 0.3 | 3.3 ± 0.9 |
| 8 | 162250 | 23 | 4.7 | 4.1 ± 0.4 | 2.7 ± 0.5 |
| 9 | 145725 | 15 | 3.1 | 3.0 ± 0.7 | 1.7 ± 0.5 |
| 10 | 893375 | 5 | 0.4 | 0.0 ± 0.2 | 0.0 ± 0.8 |
| 11 | 252625 | 7 | 1 | 0.4 ± 0.4 | 0.3 ± 0.4 |
| 12 | 722525 | 23 | 2.1 | 1.8 ± 0.4 | 5.6 ± 1.3 |
| 13 | 1108875 | 23 | 2.1 | -0.7 ± 0.4 | -8.2 ± 2.0 |
| 14 | 711675 | 21 | 2.1 | -1.5 ± 0.2 | -4.7 ± 1.0 |
| 15 | 123675 | 25 | 5.3 | -6.2 ± 1.5 | -2.9 ± 0.8 |
| 16 | 258825 | 5 | 0.8 | -0.1 ± 0.3 | -0.1 ± 0.3 |
| 17 | 1769850 | 5 | 0.3 | 0.3 ± 0.1 | 2.9 ± 0.8 |
| 18 | 253975 | 13 | 3.4 | 3.9 ± 0.7 | 9.3 ± 1.5 |
| 19 | 358525 | 14 | 1.6 | -2.1 ± 0.3 | -1.6 ± 0.8 |
| 20 | 177625 | 79 | 12.9 | -11.6 ± 1.0 | -16.4 ± 4.0 |
| 21 | 210600 | 56 | 8.5 | -27.2 ± 0.7 | -46.1 ± 7.2 |
| 22 | 208025 | 43 | 6.4 | -17.2 ± 0.8 | -28.4 ± 6.2 |
| 23 | 73925 | 99 | 18.9 | -2.9 ± 1.3 | -4.3 ± 2.1 |
| 24 | 99525 | 92 | 17.1 | 5.1 ± 1.8 | 0.2 ± 1.5 |
| 25 | 35425 | 202 | 40.5 | -1.7 ± 2.9 | -1.5 ± 1.3 |
| 26 | 41500 | 109 | 23.4 | 5.4 ± 5.2 | |
| 27 | 51275 | 44 | 7.9 | -2.0 ± 2.3 | |
| EAIS | 9909800 | 12 | 0.5 | 0.6 ± 0.1 | 16.3 ± 5.5 |
| WAIS | 1748200 | 37 | 2.8 | -5.6 ± 0.3 | -81.9 ±12.1 |
| AP | 227725 | 101 | 16.6 | 2.7 ± 2.0 | |
| | • | l | | | I |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |





Figure 1. (top) Average rate of elevation change in AIS drainage basins (Zwally et al., 2012) between 1992 and 2017 derived from 90 alternative processing scenarios. The optimal scenario is represented by a thick black line, with its estimated 1σ , 2σ and 3σ uncertainty range shaded in dark, mid, and light grey, respectively. In most basins, the average elevation rate and the spread among scenarios is close to zero. Basins showing the greatest spread among scenarios are either sparsely sampled (15 and 24 to 27) or include changes in ice thickness that are large by comparison to the spread (20 to 22). (bottom) Evaluation of the alternative processing scenarios; (i) in producing average ice sheet elevation rates (top three rows, see also Figure S9), (ii) in relation to their difference to precise airborne laser altimetry (Studinger, 2014)(bottom three rows, see also Figure S11). Against these metrics, scenario 4 (thick border) is identified to be the optimal elevation change solution.

T C C C



Figure 2A.



Figure 2. Average rate of AIS elevation change between 1992-2017 (top) and within successive 5-year intervals (bottom) from satellite radar altimetry. Black circles at the pole indicate the southern limit of the CryoSat-2 (dashed) and other (solid) satellite orbits. Grey boundaries show glacier drainage basins (Zwally et al., 2012). Green boundaries show areas of dynamical imbalance in 2017 (dark), 2007 (mid), and 1997 (light). In basins 21 and 22, the area of dynamical imbalance evolves over time.

Accept



Figure 3. Mass change and sea level contribution of areas in a state of ice dynamical imbalance (see Figure 2 for locations) and their estimated 1σ uncertainty (shaded area).